Author’s response to reviews

Title: The burden of prolonged indwelling catheter after acute urinary retention in Ikeja - Lagos, Nigeria

Authors:

Stephen O Ikuerowo (kerowq@yahoo.com)
A A Ogunade (forever@yahoo.com)
Taiwo O Ogunlowo (drogunlowo@yahoo.co.uk)
Charles C Uzodenma (parkason@doctor.com)
Julius O Esho (olusanmiesho@yahoo.com)

Version: 4 Date: 9 August 2007

Author’s response to reviews:

Dear Editor,

Thank you for accepting in principle to publish our article. We have carefully read through and make all the necessary corrections. The correction and the clean copy versions have been included.

Thank you